Author's response to reviews

Title: Prevalence of the use of cancer related self-tests by members of the public: a community survey

Authors:

Sue Wilson (s.wilson@bham.ac.uk)
Sheila Greenfield (s.m.greenfield@bham.ac.uk)
Helen M Pattison (h.m.pattison@aston.ac.uk)
Angela Ryan (a.v.ryan@bham.ac.uk)
Richard J McManus (r.j.mcmanus@bham.ac.uk)
David Fitzmaurice (d.a.fitzmaurice@bham.ac.uk)
John Marriott (j.f.marriott@aston.ac.uk)
Cyril Chapman (cyril.chapman@bwhct.nhs.uk)
Sue Clifford (s.clifford@bham.ac.uk)

Version: 2 Date: 22 August 2006

Author's response to reviews:

We have made the formatting changes requested: the list of authors in the manuscript is now the same as in the submission system; superscript numbers are now used to link the authors to their institution, and we have included the country of each institution; we have removed the keywords section; websites are now cited in the reference list; and each reference now contains all the authors names. We have also taken this opportunity to make minor amendments to the text of the manuscript.